\begin{table}%t2
%\centering
\par
\caption {
\label{tab2} 
Sources of data for molecular line opacities.
}
\begin{tabular}{l l}
\hline\hline
Species  & Reference \\
\hline
HCN vib-rot & Harris et~al. (\cite{Harris02}),\\ &J\o rgensen et~al. (2001) \\
H$_2$O vib-rot & Barber et~al. (\cite{Barber06}) \\ 
C$_2$ Phillips, Swan, \\
~~~~Ballik-Ramsay & Querci et~al. (\cite{Querci71}, priv. comm.)\\
C$_3$ & J\o rgensen et~al. (\cite{Jorgensen89})\\
C$_2$H$_2$ & J\o rgensen (\cite{Jorgensen89})\\
CH vib-rot & J\o rgensen et~al. (\cite{Jorgensen96})\\
CH A-X, B-X, C-X & Plez et~al. (\cite{Plez08})\\
CN A-X, B-X & Plez (unpublished)\\
CO vib-rot & Goorvitch (\cite{Goorvitch94}) \\
CO A-X & Kurucz (\cite{Kurucz95}) \\
CaH A-X, B-X & Plez (unpublished) \\
FeH  F$^4\Delta$-X$^4\Delta$ & Plez (unpublished) \\
MgH A-X, B$^\prime$-X & Kurucz (\cite{Kurucz95}a) \\
NH A-X & Kurucz (\cite{Kurucz95}a) \\
OH vib-rot  & Goldman et~al. (\cite{Goldman98})  \\
OH A-X & Kurucz  (\cite{Kurucz95}a) \\
SiH A-X & Kurucz  (\cite{Kurucz95}a) \\
SiO vib-rot  & Langhoff \& Bauschlicher (\cite{Langhoff93}) \\
TiO $\alpha, \beta, \gamma, \gamma^\prime, \delta, \epsilon, \phi,$ \\
   ~~~~E$^3\Pi$-B$^3\Pi$, a$^1\Delta$-f$^1\Delta$ &Plez (\cite{Plez98})\\
VO A-X, B-X, C-X & Plez (unpublished)\\
ZrO B$^1\Pi$-A$^1\Delta$, B$^1\Pi$-X$^1\Sigma$, \\ ~~~~C$^1\Sigma$-X$^1\Sigma$
      E$^1\Phi$-A$^1\Delta$, \\ ~~~~b$^3\Pi$-a$^3\Delta$, d$^3\Phi$-a$^3\Delta$,\\
      ~~~~e$^3\Pi$-a$^3\Delta$, f$^3\Delta$-a$^3\Delta$,  &Plez et~al. (\cite{Plez03}) \\
\hline
\par
\end{tabular}
\end{table}